Author’s response to reviews

Title: Effect of tolvaptan on renal handling of water and sodium, GFR and central hemodynamics in autosomal dominant polycystic kidney disease during inhibition of the nitric oxide system: a randomized, placebo-controlled, double blind, crossover study

Authors:

Safa Therwani (safatherwani@hotmail.com)

My Malmberg (mymalmbe@rm.dk)

Jeppe Rosenbaek (jepros@rm.dk)

Jesper Bech (jesper.noergaard.bech@verst.rm.dk)

Erling Pedersen (mail@ebped.dk)

Version: 2 Date: 31 Jul 2017

Author’s response to reviews: